# communications medicine

### ARTICLE

https://doi.org/10.1038/s43856-022-00217-1

OPEN

# Whole-muscle fat analysis identifies distal muscle end as disease initiation site in facioscapulohumeral muscular dystrophy

Linda Heskamp<sup>™</sup>, Augustin Ogier<sup>2</sup>, David Bendahan<sup>2</sup> & Arend Heerschap<sup>™</sup>

#### Abstract

**Background** Facioscapulohumeral dystrophy (FSHD) is a major muscular dystrophy characterized by asymmetric fatty replacement of muscles. We aimed to determine the initiation site and progression profile of the disease in lower extremity muscles of FSHD patients by assessing fat infiltration along their full proximo-distal axis using quantitative MRI.

**Methods** Nine patients underwent MRI of lower extremities to assess end-to-end muscle fat fractions (FFs) and inflammatory lesions. Seven patients underwent the same MRI ~3.5 years later. Individual muscles (n = 396) were semi-automatically segmented to calculate average FFs over all slices covering whole muscles. To assess disease progression we determined FF changes in 5 adjacent muscle segments.

**Results** We provide evidence that fat replacement commonly starts at the distal end of affected muscles where the highest FFs occur (p < 0.001). It progresses in a wave-like manner in the proximal direction at an increasing rate with the highest value ( $4.9 \pm 2.7\%$ /year) for muscles with baseline FFs of 30-40%. Thereafter it proceeds at a slower pace towards the proximal muscle end. In early phases of disease, inflammatory lesions preferentially occur at the distal muscle end. Compared with whole-muscle analysis, the common FF assessments using only few MR slices centrally placed in muscles are significantly biased (~50% in progression rate).

**Conclusions** These findings identify the distal end of leg muscles as a prime location for disease initiation in FSHD and demonstrate a wave-like progression towards the proximal end, consistent with proposed disease mechanisms. End-to-end whole-muscle fat assessment is essential to properly diagnose FSHD and its progression.

#### Plain language summary

Infiltration of fat in muscle is a feature of muscular diseases. One example is facioscapulohumeral muscular dystrophy. Here, we investigated where fat infiltration starts and how it progresses in leg muscles of patients with this disorder. We used magnetic resonance imaging to visualise the fat content of all the leg muscles. This showed that in nearly all affected muscles, fat infiltration begins in the muscles' extreme lower end, which means that disease starts at this end. Subsequently, fat infiltration progresses as a wave towards the muscle's upper end. Our observations also suggest that assessing fat content in whole muscle, rather than the common approach of only assessing the middle part of muscles, measures fat infiltration more accurately. These findings are relevant to identify factors involved in disease onset, to develop and evaluate therapies, and in disease diagnosis.





acioscapulohumeral dystrophy (FSHD) is the second most common type of muscular dystrophy worldwide. The disease is characterised by progressive skeletal muscle weakness, with the first signs often observed in the upper arm, shoulder, and face muscles, followed by the trunk and lower extremity muscles<sup>1,2</sup>.

The genetic cause of FSHD is a contraction of the D4Z4 repeat in the subtelomeric region of chromosome 4q, which releases suppression of the *DUX4* (Double HomeoBox 4) gene. Subsequently, an unknown initiation process causes expression of the DUX4 transcription factor followed by a complex cascade of events that eventually leads to muscle cell death<sup>3,4</sup>. The search for new treatments for FSHD focuses on this genetic abnormality and the suppression of DUX4 or its targets<sup>5,6</sup>. However, information about disease initiation and progression are of paramount importance for the development and evaluation of successful treatments<sup>7</sup>.

The progressive loss of muscle strength in FSHD is strongly related to the replacement of muscle tissue with fat and muscle cell atrophy<sup>8–11</sup>. MRI is an ideal tool for the investigation of these pathological processes as it can quantitatively assess the relative amount of fat in skeletal muscle tissue<sup>12</sup>. MR studies conducted on FSHD patients have revealed that some muscles are more susceptible to fat infiltration than others<sup>8-10,13-15</sup>. Furthermore, several of these studies hint towards a higher fat content in the distal direction of affected muscles, suggesting the presence of a linear gradient of increasing, proximo-distal fat content<sup>10,13,16,17</sup>. However, these MRI studies were performed on a limited number of transversal slices located at the muscle belly; therefore, it is unknown whether this proximo-distal intra-muscular fat gradient extends over the whole muscle or displays, for example, a U-shape with the highest fat infiltration at both muscle ends. The profile of this fat distribution obviously has consequences for hypotheses about disease initiation in the muscles of FSHD patients.

For these reasons, the first aim of our study was to perform a quantitative cross-sectional MR evaluation of fat infiltration covering the whole length of diseased muscles, instead of only a central section. It has been proposed that DUX4 molecules, generated only in a few myonuclei, can activate target genes in nearby nuclei and cause a transcriptional cascade of DUX4mediated dysregulation that travels along the whole muscle<sup>7,18,19</sup>. Therefore, we hypothesised that fat infiltrates diseased muscles more as a wave instead of gradually increasing along a linear gradient over the length of the muscles. To test this hypothesis, our second aim was to evaluate the progression of fat infiltration using quantitative MRI along the whole length of the proximodistal axis in muscles of FSHD patients over a period of about 3.5 years. This evaluation required the complete end-to-end analysis of the muscles using a large number of slices. To overcome the high workload of muscle delineation in all these slices, we used a semi-automatic segmentation tool<sup>20</sup>.

We detect that in nearly all affected muscles, fat infiltration starts at the distal end and progresses as a wave towards the proximal end of the muscles. We also observe that the common image analyses of fat in only the middle part of muscles may inaccurately assess fat infiltration.

#### Methods

**Study design and participants.** Nine FSHD patients were recruited for this study. Exclusion criteria were contraindications for MRI scanning, including claustrophobia, an implanted pacemaker, or being unable to lie supine for 60 min. After the patients completed their first MR exam, they were invited for a follow-up MRI 3.5 years later.

This study was conducted according to the principles of the Declaration of Helsinki (updated October 2013) and in accordance with the Medical Research Involving Human Subjects Act (WMO). The study was approved by the local medical ethics committee CMO Arnhem-Nijmegen (IRB approval number: NL66071.091.18) and all participants gave written informed consent.

MRI acquisition. All participants underwent an MRI examination of their lower extremity muscles using 3 T MR systems (Siemens, Erlangen, Germany) with a spine coil and phased array coils placed around the lower extremities. Axial 3D 2-pt Dixon images were acquired to quantify muscle fat infiltration. The lower extremity was covered from hip to ankle using 3 or 4 stacks. The most proximal stack was placed with the proximal slice at the head of the femur and each subsequent stack was placed distal to the former stack with a 45 mm overlap. Baseline MRI scans were performed on a Siemens Tim TRIO with the following settings: repetition time = 10 ms, echo times in-phase/out-phase = 2.45/3.68 ms, flip angle = 3°, field of view =  $435 \times 271$  mm, in-plane resolution =  $1.36 \times 1.36$  mm, slice thickness = 5 mm, number of slices = 72). Due to an upgrade of the MR system between the initial scan and follow-up, the followup MRI examinations were acquired on a Siemens Prisma with similar settings, except that the out-phase/in-phase echo times were 1.26 and 2.49 ms, respectively. Furthermore, inflammatory lesions were assessed with a Turbo-Inversion Recovery Magnitude (TIRM) sequence using the following settings at both baseline and followup: repetition time = 4140 ms, echo time = 41 ms, inversion time = 240 ms, flip angle =  $150^{\circ}$ , field of view =  $435 \times 271$  mm, inplane resolution =  $1.36 \times 1.36$  mm, slice thickness = 5 mm, and slice  $gap = 5 \text{ mm}^{21}$ . The subsequent stacks were aligned in the feethead direction, with no overlap.

**MRI analysis.** MR data were analysed using Matlab version 2018a (Mathworks, Natick, MA, USA). A fat fraction map was calculated from the Dixon fat and water images reconstructed by Siemens Syngo Via software (fat image/[fat image + water image]). The resulting values ranged from 0 to 100% fat. Because the MR system upgrade changed the Dixon reconstruction software, we used the Dixon images from four FSHD patients examined just before and after the upgrade to correct the baseline fat fraction maps. The relationship between the fat fraction before and after the upgrade fit a linear model (baseline fat fraction corrected =  $1.1864 \times \text{baseline}$  fat fraction - 2.7878;  $r^2 = 0.994$ ).

Subsequently, we combined the separate stacks into a single dataset covering all lower extremity muscles from hip to ankle. Baseline and follow-up scans were aligned in the feet-to-head direction to ensure both had the same proximo-distal coverage. For every leg, 12 upper leg and 10 lower leg muscles were manually segmented on every fifth slice by an experienced observer (LH). These segmentations were used to automatically segment the remaining slices by propagation (Fig. 1)<sup>20,22</sup>. This semi-automatic segmentation was verified by the experienced observer and manually corrected where needed.

The segmentations of individual muscles were used to calculate the average fat fraction of the whole muscle, including all slices, both for baseline and follow-up. Subsequently, a composite fat fraction of the whole lower extremity of each participant was estimated by taking the weighted average of the fat fractions of the individual muscles, i.e. the fat fraction of each muscle was normalised by the volume of that muscle; thus, larger muscles have more weight than smaller muscles.

Similarly, weighted composite fat fractions were calculated for the upper leg, lower leg, and for several functional muscle groups. The change in a fat fraction over time was then calculated for individual muscles and for composite scores, and was expressed as



**Fig. 1 Overview of manual and semi-automatic segmentations of lower extremity muscles displayed on fat fraction maps.** Each muscle was manually segmented at every fifth measured slice from its distal to the proximal end (**a**). The slices were then propagated, as published by Ogier et al. (2017), to segment all remaining slices (**b**). As a result, all muscles were segmented on every slice (every 5 mm). The colours represent the different muscles. Upper leg muscles: AM adductor magnus, BFL biceps femoris long head, G gracilis, RF rectus femoris, S sartorius, SM semimembranosus, ST semitendinosus, VI vastus intermedius, VL vastus lateralis, VM vastus medialis. Lower leg muscles: EDL extensor digitorum longus, FHL flexor hallucis longus, GM gastrocnemius medialis, GL gastrocnemius lateralis, P peroneus, SOL soleus, TA tibialis anterior, TP tibialis posterior.

fat fraction change (follow-up fat fraction – baseline fat fraction) and fat fraction change per year (follow-up fat fraction – baseline fat fraction/follow-up time in years).

To evaluate fat infiltration along the proximo-distal axis, we also calculated the fat fraction per slice for each muscle. Because muscle length and, therefore, the number of segmented slices differed between muscles and participants, slice numbers were expressed as a percentage of the muscle length, with the most distal slice set at 0% and the most proximal slice set at 100%. Furthermore, we divided each muscle into five equally spaced proximo-distal segments, again relative to the muscle length, and calculated the average fat fraction per segment. For example, the most distal 1/5th of the muscle was set at 0–20% and the most proximal 1/5th of the muscle was set at 80–100% of its length.

The TIRM images were reviewed by an experienced musculoskeletal radiologist and scored for the presence of signal hyperintensity. To determine at which position along the muscle length the TIRM hyperintensity preferentially occurred, we visually scored the location of the TIRM-positive lesions for all TIRM-positive muscles with low or mild fat infiltration (<30% fat fraction). For this purpose, each muscle was evenly divided into three parts: distal, middle and proximal.

**Statistics**. Statistical analysis was performed using IBM SPSS Statistics (version 25, Chicago, IL, USA). We applied several linear mixed models to manage the nesting of muscles in patients (muscle data were level 1 and patient data were level 2). The covariance structure was set at variance components and the models were fitted with a restricted maximum likelihood estimation.

The proximo-distal fat infiltration at baseline was assessed with a linear mixed model, with segment location as a fixed covariate (i.e., fat fraction =  $\beta_0 + \beta_1 \times$  segment) and intercept modelled as a random effect. This analysis was performed for all muscles and by categorising muscles based on the Mercuri scoring system into low fat-infiltrated muscles (<10% fat), mildly fat-infiltrated muscles (10–30% fat), moderately fat-infiltrated muscles (30–60% fat), and severely fat-infiltrated muscles (>60% fat)<sup>23</sup>.

The longitudinal change in fat fraction at the participant level was evaluated with a two-sided Wilcoxon signed rank test. The effect of the baseline fat fraction on this longitudinal change was calculated by dividing the muscles into groups based on their baseline fat fraction (0-10%, 10-20%, 20-30%, 30-40%, 40-50%, 50-60%, 60-70%, 70-80%, and 80-90%). We then applied a linear mixed model to compare the change in a fat fraction in each group to muscles with a baseline fat fraction of 0-10%. The baseline fat fraction group number was used as a fixed factor and the intercept was modelled as a random effect. We corrected for multiple testing with Bonferroni (number of tests = 8).

In quantitative MR studies of FSHD patients, it is common to analyse only a limited number of transversal slices. However, if fat infiltration is not homogeneously distributed along the muscle, fat fractions can be under- or over-estimated. To estimate the size of this effect in FSHD, we compared the baseline fat fraction and change in fat fraction determined over the whole muscle to these values assessed over the five middle slices or over five slices equally spread along the muscle, i.e., at 10%, 30%, 50%, 70%, and 90% of the muscle length. To this end, we calculated for each muscle the error and absolute error in baseline fat fraction and change in fat fraction. We defined error as the fat fraction (or change in fat fraction) obtained from five slices minus the fat fraction (or change in fat fraction) obtained over the whole muscle, and the absolute error is the absolute value of this difference. The average error over all muscles and limits of agreement were obtained with Bland-Altman analysis. Furthermore, we tested whether the absolute error was different from 0 using a linear mixed model, with intercept modelled as a random effect and no added covariate or factor. For the linear mixed model results, the absolute error estimated by the model is reported.

Statistical significance was set at p < 0.05. The results of the linear mixed models are displayed as estimates (standard error [SE]), and all other data were presented as mean  $\pm$  standard deviation (SD) unless stated otherwise.

**Reporting summary**. Further information on research design is available in the Nature Portfolio Reporting Summary linked to this article.

#### Results

**Participants**. We included nine patients (seven male and two female) diagnosed with FSHD type 1 and with an average age of 61 years (range: 46–72 years). Seven of these patients (seven male) underwent a follow-up MRI scan 3 years and 8 months

	Baseline fat fraction (%) (Mean ± SD; <i>n</i> = 9)	Change in fat fraction (Mean ± SD; <i>n</i> = 7)	Change in fat fraction per year (Mean ± SD; <i>n</i> = 7)
Lower extremity	32.1 ± 16.2	4.7 ± 5.9	1.3 ± 1.6
Upper legs	36.0 ± 17.8	4.7 ± 6.6	1.3 ± 1.8
Anterior compartments	27.2 ± 21.0	2.8 ± 5.1	0.8 ± 1.4
Posterior compartments	57.7 ± 20.7	4.9 ± 7.6	1.3 ± 2.1
Medial compartments	39.7 ± 18.5	9.5 ± 10.6	2.5 ± 2.8
Lower legs	25.0 ± 13.6	4.1 ± 4.7	1.1 ± 1.3
Anterior compartments	27.6 ± 16.4	1.2 ± 2.0	0.3 ± 0.6
Posterior compartments	26.1 ± 16.5	4.9 ± 5.6	1.3 ± 1.5
Lateral compartments	19.1 ± 9.7	2.2 6.3	0.6 ± 1.7

 Table 1 Average baseline fat fraction and change in fat fraction observed in the facioscapulohumeral muscular dystrophy (FSHD) patients.

Fat fractions were calculated for individual muscles and subsequently averaged to calculate weighted composite scores for the lower extremity, upper leg, lower leg, and the muscle groups, both left and right leg averaged. Standard deviations for weighted composite scores in individual patients are not presented, as some muscle groups consisted of only one or two muscles. Upper leg: anterior compartment (rectus femoris, vastii muscles and sarotrius), posterior compartment (biceps femoris, semitendinosus and semimembranosus), medial compartment (adductor magnus, adductor longus and gracilis). Lower leg: anterior compartment (tibialis anterior and extensor digitorum longus), posterior compartment (gastrocnemius, soleus, tibialis posterior, flexor hallucis longus, flexor digitorum longus and popliteus), lateral compartment (peroneus). Baseline fat fraction and change in fat fraction values of individual participants can be found in the supplemental Table S1.

after the baseline scan (range: 3 years and 5 months to 3 years and 10 months).

The evaluation included a total of 386 muscles at baseline and 298 muscles at follow-up, whereby on average, 45 slices (range: 6–91 slices) were analysed per muscle. Ten lower leg muscles were excluded from further analysis because the Dixon fat-water reconstruction for the baseline MRIs resulted in a fat-water swap in the middle of one lower leg.

**Muscle fat fraction at baseline**. The average fat fraction at baseline over all muscles and patients was  $32.1 \pm 16.2\%$  (mean  $\pm$  SD), ranging from an average fat fraction over all muscles of 8.7% for the least affected patient to 52.4% for the most affected patient. In individual muscles, whole muscle fat fraction varied from 3.1 to 89.5%. The upper leg muscles showed the highest fat fractions (Table 1 and see Supplemental Table S1 for individual participant fat fraction values).

An evaluation of the fat infiltration within individual muscles revealed a declining distal-to-proximal fat content in muscles with a mild fat fraction (10–30%, n = 102) and a moderate fat fraction (30–60%, n = 68), as illustrated for the gastrocnemius medialis, vastus lateralis and biceps femoris long head (Fig. 2a–d). In nearly all these cases, the fat content along the whole muscle length appears to follow a (shifted) reversed sigmoid curve of which the slope steepness decreases at higher overall muscular fat contents. This distal-to-proximal curve of declining fat content also occurred in some muscles with an overall low-fat fraction (<10%, n = 121) and a high-fat fraction (>60%, n = 95), but in particular for the latter fat fraction group, most muscles showed no distinct lower fat content proximally.

To evaluate intra-muscular fat infiltration quantitatively, we determined the median fat content in five segments along the length of all muscles. For this evaluation, muscles were divided into four groups of increasing overall baseline muscular fat fraction (Fig. 2e-h). The median fat fraction for a given segment depended on its position along the proximo-distal axis. Including all investigated muscles, the estimated distal-to-proximal fat fraction decline was -3.4% (0.4) (estimate [SE]) per segment (p < 0.001). This effect was most prominent for muscles with a mild or moderate fat fraction at baseline (i.e., 10-30% and 30-60%), for which the distal-to-proximal fat fractions declined by -4.0% (0.3) per segment (p < 0.001) and -9.0 (0.6) per segment (p < 0.001), respectively. This distal-to-proximal decrease in fat fraction across each segment was lower for muscles with <10% fat (-0.7% [0.1] per segment) and for muscles with >60% fat  $(-2.2\% \ [0.4] \text{ per segment}) \ (p < 0.001).$ 

**Longitudinal changes in muscle fat fraction**. At the patient level, the longitudinal average increase in a fat fraction over all lower extremity muscles of all FSHD patients during the examination period was  $4.7 \pm 5.9\%$  (p = 0.043), corresponding to a yearly increase of  $1.3 \pm 1.6\%$ . The variability between patients was large and ranged from -0.2 to 4.5% per year. The largest progression rate was observed in the upper leg muscles (Table 1).

For individual muscles, the change in overall fat fraction depended highly on their baseline value. Fat infiltration progressed faster in muscles with a baseline fat fraction between 20–30% (p < 0.001), 30–40% (p < 0.001), 40–50% (p = 0.016), and 50–60% (p = 0.048), as compared with muscles with a baseline fat fraction of <10% (Fig. 3). The largest fat fraction increase per year occurred for muscles with a baseline fat fraction of 30–40% with a rate of 4.9 ± 2.7% per year.

We then assessed the time-dependent changes in the fatinfiltrating front (slope of the reversed sigmoid curve) along the proximo-distal axis. This assessment revealed that the fatinfiltrating front had progressed towards the proximal end of the muscles during the interscan period (see Fig. 4). This observation was further confirmed by a quantitative analysis of the change in fat fraction per segment in all analysed muscles (Fig. 5). This revealed that for muscles with low baseline fat fractions (0–20%), fat fraction increases in the most distal part of the muscle, but not in the proximal segments. The change in fat fraction then moves in a wave-like manner to the proximal end with the highest increase in fat fraction at baseline fat fraction of 20–40%, after which the change in fat fraction infiltration slows down until muscles have a fat fraction of more than 70% (Fig. 5 and Supplemental Fig. S1).

Exceptions to the prevailing distal-to-proximal fat patterns. Of the 386 studied muscles, 158 muscles (40.4%) could be classified as having a decreasing fat fraction in the distal-to-proximal direction. This fat infiltration pattern was best observed in the 170 mild-to-moderately fat-infiltrated muscles (fat fraction between 10-60%), as 114 of these muscles (67.1%) showed this clear decreasing fat fraction in the distal-to-proximal direction. Of the remaining 56 mild-to-moderately fat-infiltrated muscles, 28 muscles (16.5%) with an overall fat fraction below 18% showed similar fat fractions over the whole muscle length, suggesting that these muscles have a normal fat content (Supplemental Table 2). This leaves 28 muscles (16.5% of all affected muscles) that were an obvious exception to the commonly observed FF profile (Supplemental Table 3). Four of the affected muscles (one peroneus, one extensor digitorum, one vastus medialis and one rectus femoris) showed a true counter-gradient, with higher fat



**Fig. 2 Quantitative evaluation of fat fractions along the muscle length in facioscapulohumeral muscular dystrophy (FSHD) patients.** a Fat fraction displayed at each position along the gastrocnemius medialis expressed as a percentage of muscle length from the most distal segmented slice (0%) to the most proximal segmented slice (100%). Muscles from all participants were divided into four groups: fat fraction <10% (black), 10-30% (red), 30-60% (green), and >60% (light blue). b Dixon fat fraction maps showing typical examples of fat infiltration in the gastrocnemius medialis (delineated in red). Left: sagittal view of the right lower leg of a 52-year-old male showing fat infiltration at the distal muscle end (white arrow depicts fat front). Middle: coronal view of the right lower leg of a 67-year-old female. The distal half of the gastrocnemius is fully infiltrated with fat, while the proximal half shows almost no fat infiltration (the white arrow depicts the fat front). Right: coronal view of the right lower leg of a 69-year-old male exhibiting a fully fat-infiltrated gastrocnemius medialis. **c**, **d** As figure **a**, but now for two upper leg muscles, vastus lateralis (**c**) and biceps femoris long head (**d**). **e-h** Tukey boxplots (showing median, interquartile range, and outliers) of fat fractions in five segments (distal-to-proximal) along the muscle with the corresponding linear mixed model fit for muscles with baseline fat fraction <10% (**d**), 10-30% (**e**), 30-60% (**f**), and >60% (**g**); *n* is the number of included muscles.

fractions proximally (Fig. 6a). There were also some highly fatinfiltrated semitendinosus muscles (>50% fat) showing this reversed pattern; in these muscles, it was unclear whether this was a genuine result or whether it resulted from difficulties in the manual segmentation step of these small, highly fat-infiltrated muscles. In addition, the anterior muscle compartment often showed a higher fat fraction in the most proximal slices. Occasionally, an apparent U-shaped fat pattern was observed (Fig. 6b), and several intermediately fat-infiltrated muscles displayed a homogeneous fat fraction over the whole muscle (Fig. 6c). In addition, of the affected muscles, one semimembranosus, one adductor longus, one gracilis and one vastus medialis showed a medial fat bulk (Fig. 6d).

The effect of localised sampling on the estimated change in fat fraction. Fat infiltration in leg muscles is commonly assessed by MRI using only centrally placed transversal slices. To evaluate how the proximo-distal fat gradient in muscles of FSHD patients can affect these assessments for the baseline and longitudinal changes in fat fractions, we compared fat fractions calculated from MR images spanning the whole muscle with those calculated from only the central five slices (covering 2.5 cm) of each muscle and from five slices equally spread along the muscle length, i.e., a slice placed at 10%, 30%, 50%, 70%, and 90% of the muscle length. The error, i.e., the fat fraction of five slices minus the fat fraction of the whole muscle, was calculated per muscle. The significance of the absolute error, i.e., the absolute value of this difference, was tested with a linear mixed model and displayed as the estimated absolute error (SE).

For the five middle slices, the Bland–Altman analysis showed an average error of 1.4% for the baseline fat fraction and limits of agreement from -9.2 to 12% (Fig. 7a). The estimated absolute error in the baseline fat fraction was significant, being 3.7% (0.5; p = 0.001). The error was largest in muscles with an intermediate



**Fig. 3 Change in fat fraction per year of muscles in facioscapulohumeral muscular dystrophy (FSHD) patients. a** Scatterplot of baseline fat fraction vs. change in fat fraction per year. Each dot represents one muscle. **b** Tukey boxplots (showing median, interquartile range, and outliers) per 10% baseline fat fraction group, indicating a faster progression in muscles with a baseline fat fraction between 20-60% compared with muscles with a baseline fat fraction of <10%. The mean values of fat fraction change and the number of analysed muscles (*n*) are displayed in the text above the boxes. \**p* < 0.05, \*\*\**p* < 0.001.

fat fraction, where the error varied from -11.8 to +23.5%. However, when we selected five slices equally spread over the whole muscle length, the average error was reduced to 0.1% with limits of agreement from -1.5 to 1.7% (Fig. 7a). The estimated absolute error in the baseline fat fraction was reduced sevenfold, to only 0.6% (0.1; p < 0.001). With this approach, the error also becomes independent of the baseline fat fraction.

The change in fat fraction was also biased when only five central slices were assessed. Bland–Altman analysis revealed an average error of -0.1% per year and limits of agreement from -1.6 to 1.4% per year (Fig. 7b). The corresponding estimated absolute error was 0.6% per year (0.08; p = 0.001). This effect was again most prominent in intermediately fat-infiltrated muscles, where the maximum absolute error was 3.2% per year, with a large variation between individual muscles (Fig. 7c). On average, the change in fat fraction was overestimated for muscles with a baseline fat fraction ranging from 20 to 40% and underestimated for muscles with a larger baseline fat fraction (i.e., ranging from 40 to 60%) (Fig. 7c, d). These errors in the change in fat fraction estimation were reduced when five slices evenly distributed over the whole muscle length were analysed (Fig. 7b–d). This leads to an average error of 0.0% per year and limits of agreement from

-0.5% per year to 0.5% per year, with errors being independent of baseline fat fraction. The estimated absolute error was 0.2% per year (0.03; p = 0.001).

**TIRM hyperintensity**. Of the 396 muscles studied at baseline by TIRM imaging, 215 muscles (54.3%) were scored as TIRM positive. To assess at what muscle position these TIRM lesions occurred, the location of the lesions was visually assessed for all TIRM-positive muscles with a baseline fat fraction <30% fat (n = 92 muscles). This revealed that distinct TIRM lesions were most commonly present at the distal part of these muscle (59.8%) or hyperintensity lesions occurred more diffuse from the distal end extending into the whole muscle (23.7%) (Fig. 8). Only 11.3% of the distinct TIRM-positive lesions occurred at the middle part of the muscles, and 3.1% at the proximal muscle end, while the remaining 2.1% occurred at both the distal and proximal muscle end with apparent normal signal intensities seen at the muscle belly.

Furthermore, it was observed that TIRM-positive muscles are more likely to present with a distal-proximal declining fat content compared to TIRM-negative muscles (59.9% of 212 muscles vs. 23.0% of 174 muscles). Most muscles with a lower than 10% fat fraction showed no TIRM lesions (55.2%) and they contain only 16.0% of the TIRM-positive lesions.

For the follow-up MR examinations, 308 muscles were examined, of which 181 muscles (58.7%) were TIRM positive at baseline and 194 muscles (63.0%) were TIRM positive at follow-up. Most of these muscles were TIRM positive at both time points (n = 163, 90.1%).

#### Discussion

In this MRI investigation of lower extremity muscles in FSHD patients, we quantitively assessed muscular fat infiltration in whole muscles from the distal to the proximal end, instead of only the central part of the muscle, as is common in quantitative MRI studies of diseased skeletal muscles. In this way, we demonstrate that, with a few exceptions, fat levels were highest at the distal end and lowest at the proximal end of the affected muscles. In particular, in muscles with an overall fat content between about 10 and 60%, the pattern of fat infiltration occurred as a (shifted) reversed sigmoid curve for which the slope (fat-infiltrating front) is more shallow at the higher fat content. Our longitudinal study revealed that these fat-infiltrating fronts moved to the proximal end of the muscle over the 3.5-year follow-up period. The fastest progression in the overall fat fraction was observed for muscles with mild-to-moderate fat infiltration, reaching a maximum rate of 4.9% per year in muscles with a 30-40% baseline fat fraction. Considering that muscular fat replacement is the ultimate result of cell death due to the expression of DUX4, we conclude that the disease process in FSHD starts at the distal end of leg muscles and proceeds at variable speed towards the proximal end. This conclusion is corroborated by our finding that TIRM-positive lesions, which are associated with disease activity<sup>10,17,21,24</sup>, preferentially occur at the distal end of affected muscles in the early stage of disease development in lower extremities of FSHD patients.

An important consequence of the results of our whole-muscle analysis is that the common practice in quantitative muscle MRI of assessing only a few slices at the centre of muscles may lead to large under- or overestimations of the fat fraction and its progression. In FSHD, this under- or overestimation varied from -11.8 to +23.8% for the baseline fraction and reached up to 3.4% for the fat fraction progression rate. This error can be minimised by analysing multiple transversal slices (5 or more) evenly distributed over the whole length of the muscles. In our study, semi-automatic muscle segmentation was essential for an efficient analysis of the whole muscles<sup>20</sup>.



**Fig. 4 Typical examples of the proximo-distal progression of the fat-infiltrating front in muscles of facioscapulohumeral muscular dystrophy (FSHD) patients. a** Axial and sagittal cross-sections of the right upper leg of a 47-year-old male FSHD patient showing fat fraction maps at baseline and obtained after 3.8 years follow-up. They illustrate the progression of fat infiltration from distal-to-proximal in the sartorius (red arrows) and vastus intermedius (blue arrow). The position of the fat-infiltrating front for the sartorius is depicted with a white arrow in the sagittal view for baseline and follow-up. **b** Graphical display of quantitative fat fraction curve along the sartorius displayed in figure a for baseline (solid, red) and 3.8 years later (dashed, blue). **c** Sagittal view of fat fraction maps of the leg of a 52-year-old male FSHD patient at baseline and after ~3.8 years. The white arrow indicates the fat front in the right gastrocnemius medialis. **d** Graphical display of quantitative fat fraction curve along the gastrocnemius medialis displayed in figure **c** at baseline (solid, red line) and 3.8 years later (dashed, blue). **e** Sagittal view of fat fraction maps of the upper leg of a 66-year-old male FSHD patient at baseline and 3.7 years later. The white arrow indicates the fat front in the right biceps femoris long head. **f** Graphical display of quantitative fat fraction curve along the biceps femoris long head displayed in figure **e** at baseline (solid, red) and 3.7 years later (dashed, blue).

To the best of our knowledge, this study is the first to quantitatively follow muscular fat fraction in FSHD patients along the whole muscle length, from the distal to the proximal end. The results of the cross-sectional part of this study, showing that affected muscles in the lower extremities of FSHD patients had higher fat fractions in the distal than proximal direction, are in line with earlier studies using quantitative T2 MRI for fat fraction assessment<sup>10,16</sup> and reports using DIXON MRI for this assessment<sup>13,17</sup>. However, all these studies did not assess whole muscles end-to-end, so it remained unclear whether the proximal end of the muscle also had a relatively high-fat content and whether high-fat levels extended to the far distal end in the diseased state. Our data conclusively demonstrates that the fat fraction at the proximal muscle end is not higher than in more distal muscle parts. The findings strongly suggest that fat infiltration appears with a profile resembling a (shifted) reversed sigmoid curve and proceeds in a wave-like manner with variable speed from distal-to-proximal, with a slope (fat front) becoming more shallow during disease progression. These findings contrast previous suggestions that fat infiltration occurs via a quasi-linear gradient along the length of the muscle, a conclusion which can now be ascribed to the limited muscle coverage evaluated in these studies<sup>10,16</sup>. The occurrence of high-fat fractions starting distally and extending proximally during disease progression is typical for

FSHD and differs from Duchenne muscular dystrophy, for which the highest fat fractions in the lower extremity muscles are found near both the distal and proximal ends, and the lowest at the muscle belly<sup>25</sup>.

It is remarkable that the majority of affected lower extremity muscles evaluated in the present study showed the typical pattern of fat infiltration, as described above. This implies that the far distal end of leg muscles in FSHD patients is a prime location for disease initiation, which is further supported by the preferential distal occurrence of TIRM lesions early in the disease development of a muscle. One might wonder why the distal end of these muscles is more susceptible to disease initiation than more proximal parts despite the wide variation in anatomy and function between muscles. The progression of fat infiltration also raises questions about processes underlying the spread of the disease towards the proximal end of the muscle.

Recent studies indicated that, after initial bursts of DUX4 expression in a sentinel myonucleus, the DUX4 proteins diffuse into neighbouring myonuclei and participate in the activation of DUX4 target genes<sup>7,18,19</sup>. In turn, the proteins of these genes activate further transcriptional factors leading to a transcriptional cascade, which is ultimately cytotoxic for myotubes. This gradually progresses along the whole muscle, consistent with a moving fat front. However, what initiates DUX4 bursts is still



**Fig. 5 Wave-like progression of fat infiltration along the proximo-distal muscle axis in facioscapulohumeral muscular dystrophy (FSHD) patients.** Each panel represents muscles with a 10% incremental range of baseline fat fraction (FF) values. At a baseline fat fraction <10%, only a marginal change in the average fat fraction is seen for the first most distal segment. At a baseline fat fraction of 10–20%, the average fat fraction increases in the distal segments (segments 1 and 2). At a baseline fat fraction of 20–40%, the average fat fraction increases in all segments, but most in the middle three segments (segments 2, 3, and 4). At a baseline fat fraction of 40–80%, the average fat fraction increases, especially in the proximal segments (segments 4 and 5), but the increase is lower than that for fat fraction between 20–40% and steadily diminishes at a higher baseline fat fraction. *n* is the number of included muscles. Data were presented as mean ± SD and individual data points were displayed as grey dots. Fat fraction changes normalised to residual muscle mass are displayed in the supplemental note.



Fig. 6 Typical examples of muscles in facioscapulohumeral muscular dystrophy (FSHD) patients not showing a decreasing proximo-distal fat infiltration profile. a One vastus medialis (light grey), one rectus femoris (medium grey), one peroneus (dark grey), and one extensor digitorum (black) muscles showing a reversed gradient, with a higher fat fraction proximally. b Tibialis anterior muscles showing a higher fat fraction distally and proximally.
c Muscles with a homogenously elevated fat fraction along the muscle: peroneus (light grey), vastus medialis (dark grey), and biceps femoris short head (black). d Muscles with fat infiltration in the middle of the muscle: gracilis (light grey), vastus medialis (dark grey), and rectus femoris (black).

unknown, although studies on myoblasts from FSHD patients suggest that these bursts occur stochastically<sup>26,27</sup>. Our results indicate that specific conditions at the distal end of leg muscles are involved and, thus, that initiation is not purely stochastic. In Duchenne muscular dystrophy, it has been suggested that mechanical strain on the tendons contributes to disease

initiation<sup>25</sup>, but in boys with this dystrophy, fat infiltrates muscles at both ends, in contrast to what we observed in muscles of FSHD patients.

The physiology of muscles may vary substantially along their length. Recently, the tibialis anterior was shown to have a relatively low oxidative capacity and post-exercise  $O_2$  supply in the



**Fig. 7 Effect of localised sampling on the estimated fat fraction and its progression in muscles of facioscapulohumeral muscular dystrophy (FSHD) patients. a** Bland-Altman graph displaying the error ( $\Delta$ ) in baseline fat fraction due to localised sampling of five slices in the middle (left) and spread along the muscle length (right) against the baseline fat fraction of the whole muscle (error = fat fraction of five slices - fat fraction of all slices). Each dot/square represents one muscle. Average error and limits of agreement are indicated by the red line and dashed lines, respectively. **b** Bland-Altman graph displaying the error in a change in fat fraction per year due to localised sampling of five slices - the middle (left) and spread along the muscle length (right) against the change in fat fraction in the whole muscle (error = change fat fraction of five slices - change fat fraction of all slices). **c** Error in change in fat fraction displayed against the baseline fat fraction of the whole muscle. **d** Tukey boxplots (showing median, interquartile range, and outliers) of the error (white) and absolute error (grey) for individual muscles, with the muscles grouped based on whole-muscle baseline fat fraction (10% increments). When analysing only the five central slices (left), the error (grey bars) shows that the change in fat fraction is generally overestimated in muscles with a baseline fat fraction of 20-40% and underestimated in muscles with a baseline fat fraction of 40-60%. The white bars show that in intermediately fat-infiltrated muscles (10-60% baseline fat fraction), the absolute error was about 0.8%/year. This absolute error is significantly reduced when the five slices are spread along the muscle length (right), to only 0.2%/year. *n* is the number of analysed muscles.



**Fig. 8 Typical examples of hyperintensity patterns along the muscle length at a turbo-inversion recovery magnitude (TIRM) sequence.** The order of the images from distal-to-proximal is indicated at the top of the figure. **a** Lower legs of a 52-year-old male facioscapulohumeral muscular dystrophy (FSHD) patient showing TIRM hyperintensity in the distal part of the gastrocnemius lateralis and a normal appearance of this muscle at the proximal muscle end (arrows). **b** Upper legs of a 47-year-old male FSHD patient with TIRM hyperintensity localised in the distal part of the sartorius (arrow heads) and vastus medialis (arrows) with a normal appearance at the proximal end of these muscles. **c** Lower legs of a 62-year-old male FSHD patient showing TIRM hyperintensity over the whole muscle length for all lower leg anterior compartment muscles (arrows).

distal part of the muscle<sup>28,29</sup>. This distally lower post-exercise O<sub>2</sub> supply also occurs in the quadriceps and gastrocnemius<sup>30–33</sup>. Interestingly, oxidative stress in FSHD-affected muscle cells is implicated in disease onset by increasing DUX4 expression<sup>34</sup>. Moreover, DUX4 activates oxidative stress responses, leading to cell death<sup>35–37</sup>, and affected muscle cells are vulnerable to oxidative stress<sup>38,39</sup>. Therefore, it is of interest to know if (transient) hypoxia at the distal end of FSHD-affected leg muscles plays a role in disease onset, our results provide a framework for further investigation into the nature of this process.

The overall change in the fat fraction of 1% per year indicates that fat progression in the lower extremities of FSHD patients at the individual level is slow. This percentage change is relatively low compared to those observed in other FSHD studies, which ranged from 0.86% per year to 5.4% per year<sup>10,14,15,24,40</sup>. This large variability among reported progression rates in a given patient cohort is not surprising, given that the rates are affected by multiple factors such as muscle type, the investigated location along the proximo-distal axis, the disease severity at baseline, and the applied MRI technique. For example, our study and that of others showed that fat infiltration in upper leg muscles progresses faster as compared to lower leg muscles<sup>14,15</sup>. Moreover, previous studies have assessed mainly central muscle areas, which can lead to over- or underestimation of fat progression (vide infra), which also depends on baseline fat fraction. Furthermore, more rapid fat infiltration rates have been reported in the muscles of FSHD patients involved in active disease processes, i.e., those showing TIRM-positive foci or those in a state of intermediate fat infiltration<sup>24</sup>. Finally, methodological differences, such as different MRI acquisition methods (e.g., 2 pt Dixon, multi-echo spinecho) and data processing tools, may have slightly different sensitivities in the quantification of fat fractions. For example, the 2pt-Dixon study used in our study is likely to be less accurate than a multi-point Dixon, however, it does not affect precision and this precision is what really matters in longitudinal studies.

Despite that the leg muscle fat infiltration rate at the participant level was slow, fat infiltration can progress quickly in individual muscles. The fat replacement rate rapidly accelerates up to an average of 4.9% fat per year following fat infiltration onset until a muscle's fat fraction becomes about 40%. At this stage, the fat replacement rate decelerates until muscles are completely fat-infiltrated. This typical pattern is in line with other longitudinal studies conducted in FSHD patients<sup>10,14,15</sup>. The pattern also fits the clinical picture reported in FSHD patients, with long stable periods alternating with short periods of rapid deterioration in single muscles or muscle groups<sup>41</sup>. In other muscular dystrophies, the relationship between the baseline fat fraction and the change in fat fraction has been studied in less detail. In late-onset Pompe disease, myotonic dystrophy type 1, and spinal muscular dystrophy, the change in fat fraction in muscles with elevated baseline fat fraction was larger than in muscles with fat fractions in the range of healthy controls<sup>42–44</sup>.

The presence of a distal-to-proximal fat-infiltrating front in diseased muscles has major implications for the application of quantitative MRI as a biomarker to assess treatments in clinical trials<sup>6,45,46</sup>. We demonstrated that evaluating fat fractions only in the central muscle parts, a commonly applied approach, instead of considering the whole muscle length, led to an average absolute error of 3.7% in baseline fat fraction and 0.6% per year in fat replacement rate in lower extremity muscles. The latter estimated absolute error is very large, representing 50% of the total change in fat fraction over 1 year commonly reported<sup>15,24</sup>. For instance, even if clinical trials intend to monitor disease progression in intermediately fat-infiltrated leg muscles-which are known to be most sensitive to changes-the corresponding evaluation of baseline fat fraction and fat replacement rate may both be largely biased if measurements are performed for slices located away from the fat progression front. Such an observation was made in a study of muscles in Duchenne muscular dystrophy, where a slight shift of 1.5 cm along the proximo-distal muscle axis led to a significant difference in the reported fat fraction<sup>25</sup>, especially in intermediately fat-infiltrated muscles. The absolute errors in baseline fat fraction and change in fat fraction can be substantially reduced when the five analysed slices are distributed along the muscle length. These findings demonstrate that whole-muscle

imaging is essential to properly diagnose disease severity and to assess disease progression and response to therapy, such as in clinical trials. However, given that manual segmentation in quantitative assessments of whole muscle is very time-consuming, automatic or semi-automatic muscle segmentation approaches are critical in clinical assessments<sup>47</sup>. The analysis of an adequate number of slices properly divided over the length of muscles may represent a valid alternative.

In conclusion, fatty replacement in affected lower extremity muscles of FSHD patients commonly starts at the distal end of the muscles. It progresses relatively rapidly, in a wave-like manner, in the proximal direction in the mid-phase of disease development, after which it proceeds at a slower pace towards the proximal muscle end. In the early phase of disease development, TIRMpositive lesions dominantly occur at the distal end of affected muscles. These observations identify the distal end of lower extremity muscles as a prime location for disease initiation and are consistent with proposed mechanisms for disease progression in muscles. Factors involved in the activation of DUX4 expression to initiate disease have yet to be discovered, but oxidative stress at the distal end of leg muscles may play a role. Our analysis of whole muscles in FSHD demonstrates that the common practice in muscle MRI studies of assessing only a few slices in more central parts of muscles may lead to large under- or overestimations of muscle fat fractions and their change over time.

#### Data availability

Source data for the figures are available as Supplementary Data 1. All post-processed data generated or analysed during this study are included in Supplementary Data 2. All other data were available from the corresponding author upon reasonable request.

Received: 13 February 2022; Accepted: 11 November 2022; Published online: 01 December 2022

#### References

- Tawil, R., van der Maarel, S. M. & Tapscott, S. J. Facioscapulohumeral dystrophy: the path to consensus on pathophysiology. *Skelet. Muscle* 4, 12 (2014).
- Mul, K. et al. What's in a name? The clinical features of facioscapulohumeral muscular dystrophy. *Pract. Neurol.* 16, 201–207 (2016).
- Van der Maarel, S. M., Tawil, R. & Tapscott, S. J. Facioscapulohumeral muscular dystrophy and DUX4: Breaking the silence. *Trends Mol. Med.* 17, 252–258 (2011).
- Lim, K. R. Q., Nguyen, Q. & Yokota, T. DUX4 signalling in the pathogenesis of facioscapulohumeral muscular dystrophy. *Int. J. Mol. Sci.* 21, 729 (2020).
- Schätzl, T., Kaiser, L. & Deigner, H.-P. Facioscapulohumeral muscular dystrophy: genetics, gene activation and downstream signalling with regard to recent therapeutic approaches: an update. *Orphanet J. Rare Dis.* 16, 129 (2021).
   Wang, L. H. & Tawil, R. Current therapeutic approaches in FSHD. *J.*
- Wang, L. H. & Tawii, R. Current therapeutic approaches in FSHD. J. Neuromuscul. Dis. 8, 441–451 (2021).
- Tassin, A. et al. DUX4 expression in FSHD muscle cells: how could such a rare protein cause a myopathy? J. Cell. Mol. Med. 17, 76–89 (2013).
- Olsen, D. B., Gideon, P., Jeppesen, T. D. & Vissing, J. Leg muscle involvement in facioscapulohumeral muscular dystrophy assessed by MRI. *J. Neurol.* 253, 1437–1441 (2006).
- Iosa, M. et al. Mobility assessment of patients with facioscapulohumeral dystrophy. *Clin. Biomech.* 22, 1074–1082 (2007).
- Janssen, B. H. et al. Distinct disease phases in muscles of facioscapulohumeral dystrophy patients identified by MR detected fat infiltration. *PLoS ONE* 9, e85416 (2014).
- Marra, M. A. et al. Specific muscle strength is reduced in facioscapulohumeral dystrophy: an MRI based musculoskeletal analysis. *Neuromuscul. Disord.* 28, 238–245 (2018).
- Carlier, P. G. et al. Skeletal muscle quantitative nuclear magnetic resonance imaging and spectroscopy as an outcome measure for clinical trials. *J. Neuromuscul. Dis.* 3, 1–28 (2016).
- Mul, K. et al. Adding quantitative muscle MRI to the FSHD clinical trial toolbox. *Neurology* 89, 2057–2065 (2017).

- Andersen, G. et al. MRI as outcome measure in facioscapulohumeral muscular dystrophy: 1-year follow-up of 45 patients. J. Neurol. 264, 438–447 (2017).
- Wang, L. H. et al. Longitudinal study of MRI and functional outcome measures in facioscapulohumeral muscular dystrophy. *BMC Musculoskelet*. *Disord.* 22, 262 (2021).
- Kan, H. E. et al. Quantitative MR imaging of individual muscle involvement in facioscapulohumeral muscular dystrophy. *Neuromuscul. Disord.* 19, 357–362 (2009).
- Dahlqvist, J. R. et al. Relationship between muscle inflammation and fat replacement assessed by MRI in facioscapulohumeral muscular dystrophy. J. Neurol. 0, 0 (2019).
- Feng, Q. et al. A feedback loop between nonsense-mediated decay and the retrogene DUX4 in facioscapulohumeral muscular dystrophy. *Elife* 4, 1–13 (2015).
- Rickard, A. M., Petek, L. M. & Miller, D. G. Endogenous DUX4 expression in FSHD myotubes is sufficient to cause cell death and disrupts RNA splicing and cell migration pathways. *Hum. Mol. Genet.* 24, 5901–5914 (2015).
- Ogier, A., Sdika, M., Foure, A., Le Troter, A. & Bendahan, D. Individual muscle segmentation in MR images: a 3D propagation through 2D non-linear registration approaches. In 39th Annual International Conference of the IEEE Engineering in Medicine and Biology Society (EMBC) 317-320 (IEEE, 2017).
- Frisullo, G. et al. CD8(+) T cells in facioscapulohumeral muscular dystrophy patients with inflammatory features at muscle MRI. J. Clin. Immunol. 31, 155–66 (2011).
- Ogier, A. C. et al. A novel segmentation framework dedicated to the follow-up of fat infiltration in individual muscles of patients with neuromuscular disorders. *Magn. Reson. Med.* 83, 1825–1836 (2020).
- 23. Mercuri, E. et al. Muscle MRI in inherited neuromuscular disorders: past, present, and future. J. Magn. Reson. Imaging 25, 433-440 (2007).
- Dahlqvist, J. R. et al. Evaluation of inflammatory lesions over 2 years in facioscapulohumeral muscular dystrophy. *Neurology* 95, e1211–e1221 (2020).
- Hooijmans, M. T. et al. Non-uniform muscle fat replacement along the proximodistal axis in Duchenne muscular dystrophy. *Neuromuscul. Disord.* 27, 458–464 (2017).
- Haynes, P., Bomsztyk, K. & Miller, D. G. Sporadic DUX4 expression in FSHD myocytes is associated with incomplete repression by the PRC2 complex and gain of H3K9 acetylation on the contracted D4Z4 allele. *Epigenetics Chromatin* 11, 47 (2018).
- 27. Snider, L. et al. Facioscapulohumeral dystrophy: incomplete suppression of a retrotransposed gene. *PLoS Genet.* **6**, e1001181 (2010).
- Heskamp, L. et al. Post-exercise intramuscular O 2 supply is tightly coupled with a higher proximal-to-distal ATP synthesis rate in human tibialis anterior. *J. Physiol.* 599, 1533–1550 (2021).
- 29. Boss, A. et al. Oxidative capacity varies along the length of healthy human tibialis anterior. *J. Physiol.* **596**, 1467–1483 (2018).
- Niess, F. et al. Interleaved 31 P MRS/ 1 H ASL for analysis of metabolic and functional heterogeneity along human lower leg muscles at 7T. *Magn. Reson. Med.* 83, 1909–1919 (2020).
- 31. Mizuno, M. et al. Regional differences in blood flow and oxygen consumption in resting muscle and their relationship during recovery from exhaustive exercise. J. Appl. Physiol. **95**, 2204–2210 (2003).
- Miura, H., McCully, K., Hong, L., Nioka, S. & Chance, B. Regional difference of muscle oxygen saturation and blood volume during exercise determined by near infrared imaging device. *Jpn. J. Physiol.* 51, 599–606 (2001).
- Crenshaw, A. G., Bronee, L., Krag, I. & Jensen, B. R. Oxygenation and EMG in the proximal and distal vastus lateralis during submaximal isometric knee extension. J. Sports Sci. 28, 1057–1064 (2010).
- Sasaki-Honda, M. et al. A patient-derived iPSC model revealed oxidative stress increases facioscapulohumeral muscular dystrophy-causative DUX4. *Hum. Mol. Genet.* 27, 4024–4035 (2018).
- Turki, A. et al. Functional muscle impairment in facioscapulohumeral muscular dystrophy is correlated with oxidative stress and mitochondrial dysfunction. *Free Radic. Biol. Med.* 53, 1068–1079 (2012).
- Lek, A. et al. Applying genome-wide CRISPR-Cas9 screens for therapeutic discovery in facioscapulohumeral muscular dystrophy. *Sci. Transl. Med.* 12, eaay0271 (2020).
- Karpukhina, A. et al. Analysis of genes regulated by DUX4 via oxidative stress reveals potential therapeutic targets for treatment of facioscapulohumeral dystrophy. *Redox Biol.* 43, 102008 (2021).
- Saada, Y. B. et al. Facioscapulohumeral dystrophy myoblasts efficiently repair moderate levels of oxidative DNA damage. *Histochem. Cell Biol.* 145, 475–483 (2016).
- Bosnakovski, D. et al. An isogenetic myoblast expression screen identifies DUX4-mediated FSHD-associated molecular pathologies. *EMBO J.* 27, 2766–2779 (2008).
- Fatehi, F. et al. Long-term follow-up of MRI changes in thigh muscles of patients with Facioscapulohumeral dystrophy: a quantitative study. *PLoS ONE* 12, e0183825 (2017).

## ARTICLE

- Tawil, R. & Van Der Maarel, S. M. Facioscapulohumeral muscular dystrophy. Muscle Nerve 34, 1–15 (2006).
- Carlier, P. G. et al. Skeletal muscle quantitative nuclear magnetic resonance imaging follow-up of adult Pompe patients. *J. Inherit. Metab. Dis.* 38, 565–572 (2015).
- Heskamp, L. et al. Quantitative muscle MRI depicts increased muscle mass after a behavioral change in myotonic dystrophy type 1. *Radiology* 297, 132–142 (2020).
- Otto, L. A. M. et al. Quantification of disease progression in spinal muscular atrophy with muscle MRI—a pilot study. NMR Biomed. 34, e4473 (2021).
- LoRusso, S. et al. Clinical trial readiness to solve barriers to drug development in FSHD (ReSolve): protocol of a large, international, multi-center prospective study. *BMC Neurol.* 19, 224 (2019).
- Mellion, M. L. et al. Phase 1 clinical trial of losmapimod in facioscapulohumeral dystrophy: Safety, tolerability, pharmacokinetics, and target engagement. *Br. J. Clin. Pharmacol.* 87, 4658–4669 (2021).
- Ogier, A. C., Hostin, M.-A., Bellemare, M.-E. & Bendahan, D. Overview of MR image segmentation strategies in neuromuscular disorders. *Front. Neurol.* 12, 625308 (2021).

#### Acknowledgements

This work was funded by Friends of FSH Research (awarded in 2018) and the initial phase of this project was supported by aTyr Pharma. We would like to thank our participants and the research assistants for their help in contacting the participants. Furthermore, we thank Sanne Vincenten for her help with comparing data acquired before and after the MR system upgrade and Marieke Ploegmakers for scoring the TIRM images.

#### Author contributions

The experiments were performed by L.H. at the department of Medical Imaging/Radiology of the Radboud university medical center, Nijmegen, The Netherlands. L.H. and A.H. were involved in the conception or design of the work. A.O. developed the semiautomatic segmentation software under the supervision of D.B. and applied it to the FSHD data of this study. L.H. did further data processing and wrote the first draft. All authors were involved in revising the manuscript.

#### **Competing interests**

The authors declare no competing interests.

#### **Additional information**

**Supplementary information** The online version contains supplementary material available at https://doi.org/10.1038/s43856-022-00217-1.

**Correspondence** and requests for materials should be addressed to Linda Heskamp or Arend Heerschap.

Peer review information Communications Medicine thanks the anonymous reviewers for their contribution to the peer review of this work. Peer reviewer reports are available.

Reprints and permission information is available at http://www.nature.com/reprints

**Publisher's note** Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

**Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons license, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons license and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this license, visit http://creativecommons.org/ licenses/by/4.0/.

© The Author(s) 2022